Author's response to reviews

Title: Evaluation of QUADAS, a tool for the quality assessment of diagnostic accuracy studies

Authors:

Penny F Whiting (penny.whiting@bristol.ac.uk)
Marie E Westwood (mew3@york.ac.uk)
Anne W S Rutjes (a.rutjes@amc.uva.nl)
Johannes B Reitsma (j.reitsma@amc.uva.nl)
Patrick M M Bossuyt (p.m.bossuyt@amc.uva.nl)
Jos Kleijnen (jos@kleijnen.freeserve.co.uk)

Version: 3 Date: 4 January 2006

Author's response to reviews:

Dear Editors,

Thank you for accepting our manuscript. We have made the formatting changes that you requested in this revised version of the manuscript:

Title & Abstract - these are identical in the manuscript and in the submission system.

Author list - the list of authors in the manuscript is now written exactly as they are in the submission system.

Word count - we have removed this from the manuscript.

Box 1 - We have renamed the boxes as tables (box 1 has become table 1, box 2 has become table 4) and we have renumbered the other tables accordingly (table 1 has become table 2 and table 2 has become table 3). We have also made these changes in the text.

Tables - We have removed the shading in table 1 and have instead used italics to highlight the relevant section

Best wishes,

Penny Whiting